Author’s response to reviews

Title: Posterior decompression and occipitocervical fixation followed by intraoperative vertebroplasty for metastatic involvement of the axis

Authors:

Xinjie WU (wuxinjie_ms@163.com)
Yingna QI (ayingqi119925@163.com)
mingsheng tan (doctortan_123@126.com)
Ping YI (dryiping_123@163.com)
Feng YANG (dryangfeng@126.com)
Xiangsheng TANG (drtangxiangsheng@126.com)
Qingying HAO (drhaoqingying@126.com)

Version: 3 Date: 18 Dec 2017

Author’s response to reviews:

Editor Comments:

Thanks for the lots of effort you put into this work.

The reviewers have suggested publication of this manuscript and I will follow them as soon as you have corrected the following sentence in the results section of the abstract:

"The median time was 7 months." : Do you mean survival?

Authors: Sorry for any inconvenience we have brought to you and we have revise the sentence in P2L26 which is highlighted in red.